Author’s response to reviews

Title: Deposition of respiratory virus pathogens on frequently touched surfaces at airports

Authors:
Niina Ikonen (niina.ikonen@thl.fi)
Carita Savolainen-Kopra (carita.savolainen-kopra@thl.fi)
Joanne Enstone (joanne.enstone@nottingham.ac.uk)
Ilpo Kulmala (ilpo.kulmala@vtt.fi)
Pertti Pasanen (pertti.pasanen@uef.fi)
Anniina Salmela (anniina.salmela@uef.fi)
Satu Salo (satu.salo@vtt.fi)
Jonathan Nguyen-Van-Tam (jonathan.van-tam@nottingham.ac.uk)
Petri Ruutu (petri.ruutu@thl.fi)

Version: 3 Date: 08 May 2018

Author’s response to reviews:

INFD-D-18-00431

Deposition of respiratory virus pathogens on frequently touched surfaces at airports Niina Ikonen, MSc; Carita Savolainen-Kopra, PhD; Joanne E. Enstone, PhD; Ilpo Kulmala, PhD; Pertti Pasanen, PhD; Anniina Salmela, MSc; Satu Salo, PhD; Jonathan S. Nguyen-Van-Tam, PhD; Petri Ruutu, PhD BMC Infectious Diseases

Dear Editor of BMC Infectious Diseases

We have reviewed the latest comment suggested. Below is our response to the comment.
1. Adapting an image still requires you to provide the source. Please acknowledge the source Figure 1 was adapted from in the Figure legend and include the source as a reference in the list of references.

- The figure has only been made for this manuscript. The images of an aeroplane in the figure have been drawn by one of the authors of the manuscript. The whole picture is our own and we have not adapted it from anything else. Because the picture is entirely our own, we do not have a reference for it.

We hope that you find the manuscript suitable for publication.

Yours sincerely,

Niina Ikonen